Author’s response to reviews

Title: Exploring the experiences of being an ethnic minority student within an undergraduate nursing education: A qualitative study

Authors:

Sylvi Monika Flateland (sylvi.flateland@ui.no)
Maxine Pryce-Miller (Maxine.Pryce-Miller@uwe.ac.uk)
Anne Valen-Sendstad Skisland (anne.skisland@ui.no)
Anne Flaatten Tønsberg (anne.k.tonsberg@ui.no)
Ulrika Söderhamn (ulrika.soderhamn@ui.no)

Version: 3 Date: 20 Nov 2019

Author’s response to reviews:

NURS-D-19-00015R2
Exploring the experiences of being an ethnic minority student within an undergraduate nursing education: A qualitative study
Sylvi Monika Flateland, MSc; Maxine Pryce-Miller, PhD; Anne Valen-Sendstad Skisland, MSc; Anne Flaatten Tønsberg, MSc; Ulrika Söderhamn, PhD

20 November 2019

To
Janelle Coore
BMC Nursing

I hereby send you the version with the necessary corrections of the manuscript for publication in BMC Nursing, written by Sylvi Flateland, Maxine Pryce-Miller, Anne Valen-Sendstad Skisland, Anne Kristin Flaatten Tønsberg and Ulrika Söderhamn, entitled “Exploring the experiences of being an ethnic minority student within undergraduate nurse education: A qualitative study”.

Correspondence regarding this manuscript is to Sylvi Flateland.

Sylvi Monika Flateland, Associate Professor, RN, MSc
Department of Health and Nursing Sciences
Faculty of Health and Sport Sciences
University of Agder
Post Box 422
The authors are grateful for the comments and here follows the response:

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Please note that an 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Answer: Sylvi Monika Flateland, Anne Valen-Sendstad Skisland, Anne Flaatten Tønsberg and Ulrika Söderhamn designed the study; Sylvi Flateland performed and analysed the interviews; Sylvi Flateland, Maxine Pryce-Miller, Anne Flaatten Tønsberg, Anne Valen-Sendstad Skisland and Ulrika Söderhamn discussed the analysis; Sylvi Flateland, Maxine Pryce-Miller and Ulrika Söderhamn drafted the article; all authors revised it critically and all authors read and approved the last version.

2. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Answer: There are not any abbreviations in the paper.

3. Please remove the tracked changes, and response letter documents from the file inventory as it is no longer needed at this stage.

Answer: Tracked changes are removed in the paper and other documents which is not of interest are also removed.

4. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.
Answer: The manuscript has been proofread. A clean copy of the manuscript will be uploaded.

Best wishes Sylvi Flateland